Author's response to reviews

Title: Gliomatosis cerebri presenting as rapidly progressive dementia and parkinsonism in an elderly woman: a case report

Authors:

emmanuelle duron (emmanuelle.duron@brc.aphp.fr)
anne lazareth (lazarethanne@yahoo.fr)
jean-yves gaubert (jean-yves.gaubert@brc.aphp.fr)
carole raso (carole.raso@brc.aphp.fr)
olivier hanon (olivier.hanon@brc.aphp.fr)
anne-sophie rigaud (anne-sophie.rigaud@brc.aphp.fr)

Version: 9 Date: 13 February 2008

Author's response to reviews: see over
Dr. Duron Emmanuelle  
Service de Gériatrie. Professeur Rigaud  
Hopital Broca  
54-56 rue Pascal  
75013 Paris  
France  
Tel: + 33144083503  
Fax: +33144083510  
E-mail: emmanuelle.duron@brc.aphp.fr

Pr Michael Kidd  
Editor in chief  
Journal of medical case report

13 02 2008

Dear Professor Kidd,

Please, find enclosed this manuscript by Dr Duron Emmanuelle et al. that we would like to resubmit after asked formatting changes:

“Gliomatosis cerebri presenting as rapidly progressive dementia and Parkinsonism in an elderly woman. a case report”

As suggested by the editorial team recommended revisions have been made:

- The author’s names have been written identically and in the order in which they are listed in the submission system.

- Authors’ qualification have been removed from the title page

- All the authors’ email addresses have been included on the title page with the suggested format.

- The individual author sections were removed on the title page

- Sentence case was used for all headings and subheadings.

- The consent section was moved to following the authors’ contribution

- No acknowledgement section was included and there was no source of funding for this work
- The term “et al” was removed from the references. References 4, 7, 12 and 13 were updated.

- The word “legend” was removed from the figure legends.

- The three figures have been cropped as closely as possible.

We hope that you will consider it suitable for publication. Looking forward to hearing from you.

Sincerely yours,

Dr Emmanuelle Duron, MD